RESEARCH

Open Access

A look in the mirror - body exposure in clinical practice



Jessica Werthmann^{1*}, Eva Naumann², Silja Vocks³, Jennifer Svaldi² and Andrea S. Hartmann⁴

Abstract

Background The dissemination of evidence-based techniques is critical for the successful treatment of eating disorders in clinical practice. A growing number of studies suggests that body exposure is an effective technique to treat body image disturbance in eating disorders. However, the dissemination of body exposure among psychotherapists in clinical practice remains unclear.

Methods An online survey was conducted among licensed psychotherapists in Germany. The dissemination of body exposure in clinical practice, psychotherapists' characteristics (such as clinical training, attitudes towards exposure, confidence), and therapists' experiences with benefits and side-effects of body exposure were assessed.

Results Data of 230 psychotherapists were analysed. More than half of them (58.3%) applied body exposure in their clinical practice. Yet, body exposure was only offered to 37.3% of their eating disorder patients. Moreover, 56.7% of psychotherapists delivering body exposure indicated that they have not received any training in this technique. Self-reported confidence in delivering body exposure correlated significantly with the number of patients treated with body exposure. Psychotherapists who applied body exposure reported minor side-effects and that the majority of their patients profited from this technique.

Conclusions Our results offer insights into the dissemination of body exposure in clinical practice in Germany. Overall, body exposure is still underused considering the empirical evidence demonstrating the potential of body exposure to treat body image disturbances effectively. Moreover, with regard to potential barriers of using body exposure, our data suggest that training opportunities for clinicians may facilitate the dissemination of this technique in clinical practice.

Keywords Body exposure, Exposure therapy, Cognitive-Behavioural therapy, Eating disorders, Dissemination, Evidence-based practice, Clinician anxiety, Therapist drift, Therapist's characteristics, Research-Practice gap

Plain English summary

Body exposure is an evidence-based technique to treat body image disturbances. In this study, we assessed the use of body exposure among psychotherapists in clinical practice in Germany and found that body

*Correspondence: Jessica Werthmann Werthmann@psychologie.uni-freiburg.de

Full list of author information is available at the end of the article



© The Author(s) 2025. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

exposure is still underused in the treatment of eating disorders. Our data call for more training opportunities in this area, particularly as confidence in this technique is associated with more application.

Body image disturbance is a core symptom of eating disorder (ED) pathology (American Psychiatric Association 2013; Fairburn et al. 2003; Stice and Shaw 2002). Residing body dissatisfaction contributes to relapse after treatment (Jacobi et al. 2004). Targeting body dissatisfaction is therefore an integral part of ED treatment in several treatment manuals (e.g., Legenbauer and Vocks 2014; Svaldi and Tuschen-Caffier 2018; Tuschen-Caffier and Florin 2012; Vocks et al. 2018). A meta-analysis on the effects of interventions for body dissatisfaction demonstrated that body exposure (BE) is a promising technique towards this aim (Alleva et al. 2015). In brief, BE entails the repetitive and systematic description of oneself, for example in the mirror, usually under therapeutic guidance. This technique can be used for adult and adolescent patients with body dissatisfaction, however research testing BE in adolescents is scarce (Biney et al. 2021; Hartmann et al. 2021).

Similar to exposure in anxiety disorders, different concepts underlying its efficacy have been investigated. Most prominently, these focus on habituation (Díaz-Ferrer et al. 2015; Hilbert and Tuschen-Caffier 2004; Moreno-Domínguez et al. 2012; Trentowska et al. 2013, 2017; Vocks et al. 2007b, 2008). Yet, recently alternative theoretical concepts, including attention bias modification (e.g., Glashouwer et al. 2016; Krohmer et al. 2022), cognitive change (e.g., Baur et al. 2022) and reduction of body perception distortion (e.g., Lewer et al. 2017) have been tested as well (see for an overview e.g., Hartmann et al. 2021).

BE has positive effects on several ED-related symptoms such as body dissatisfaction, negative mood, low self-esteem and dysfunctional body-related information processing (Griffen et al. 2018; Hartmann et al. 2021; Krohmer et al. 2022). However, despite its efficacy, very little is known about the use of BE in clinical practice. Notably, against recommendations in clinical guidelines for a range of mental disorders (e.g., Arbeitsgemeinschaft der Wissenschaftlichen Medizinischen Fachgesellschaften 2019; National Institute for Health and Care Excellence, 2017), exposure is generally often poorly disseminated (Becker-Haimes et al. 2017; Cook et al. 2004; Kline et al. 2021). For example, 87% of more than 2000 participating mental health clinicians in the U.S. indicated that they have never applied exposure-based techniques in their daily practice (Cook et al. 2010). Similarly, a survey of licensed psychotherapists in Germany found that less than half of the respondents used exposure when treating patients with anxiety disorders (Pittig and Hoyer 2017).

This striking gap between evidence and practice has stimulated considerable research, particularly in the field of anxiety disorders. Several factors have been identified and associated with the limited use of exposure. Most importantly, negative attitudes towards exposure and exposure-related concerns (e.g., emotional distress or the expectation that patients may not tolerate exposure) reduce the likelihood of delivering exposure in daily practice for anxiety disorders (Becker-Haimes et al. 2017; de Jong et al. 2020; Deacon et al. 2013; Kline et al. 2021; Schumacher et al. 2019). Other potentially related factors include therapist characteristics (e.g., age and experience (de Jong et al. 2020; Deacon et al. 2013; Finch et al. 2020), clinical setting (de Jong et al. 2020; Harned et al. 2013; Kline et al. 2021; Milgram et al. 2022; Pittig and Hoyer 2017), as well as therapists' misunderstanding of the exposure rationale (Kline et al. 2021). However, emerging research suggests that training opportunities in exposure may facilitate the implementation of exposure in daily practice by influencing attitudes, concerns and confidence regarding the delivery of exposure (Farrell et al. 2016; Trivasse et al. 2020; Wright and Waller 2020).

With regard to the implementation of BE in the treatment of EDs, there is a lack of evidence about the dissemination in daily practice and factors that may influence use of BE. Against this backdrop, the primary aim of this study was to assess the prevalence of applying BE among licensed psychotherapists treating patients with EDs in clinical practice in Germany. A secondary aim was to assess how psychotherapists' factors such as age, gender, clinical background, negative beliefs and attitudes towards BE are associated with the delivery of BE in ED treatments. Finally, we wanted to evaluate psychotherapists' experiences with the benefits and side-effects of BE in clinical practice.

Method

Recruitment

Participants were included if they indicated to hold a formal license as psychotherapist in Germany (i.e., "Approbation"), including adult psychotherapists, child and adolescent psychotherapists as well as medical doctors with psychotherapy training. We recruited licensed psychotherapists via their registered email at the Association of German Health Insurance Providers (i.e., the "Kassenärztliche Vereinigung"), in which all licensed psychotherapists must register. We contacted psychotherapists in several federal states of Germany (i.e., Baden-Wurttemberg, a section of North Rhine-Westphalia, Rhineland Palatinate, Brandenburg, Saxony and Schleswig-Holstein). Additionally, inpatient clinics in these federal states were hand-searched via internet and the head psychologist or the director of the clinic was contacted via email with the request to pass on the study invitation to psychotherapy-licensed employees.

This invitation email encouraged licensed psychotherapists to participate in an online survey about their experiences with BE in the treatment of EDs. The email contained a link to our electronic survey and participants were informed that they would be offered a free online workshop on BE if they completed the survey.

Materials

The online survey was programmed and distributed via the German online survey platform Unipark (Tivian XI GmbH, 2021) and consisted of 25 main questions which contained, partly, further sub-questions, see supplementary materials for an overview and Open Science Framework Project (https://osf.io/fz9ys/) for the complete German-language survey.

Overall, the survey contained questions about the dissemination of BE in daily practice, (e.g., if therapists have used BE and whether other techniques have been used to treat body image disturbances; aim 1: dissemination). It also contained questions concerning the psychotherapist's characteristics (e.g., demographics, clinical background, confidence with BE, positive attitudes and negative beliefs; aim 2: therapist factors associated with the use of BE) and, finally, there were also questions concerning potential benefits and side-effects of BE (aim 3: experiences in clinical practice). Most questions were measured with a four-point Likert scale (ranging from 1 "Not applicable at all" to 4 "Very much applicable"). Some questions required single answer selection (e.g., "Did you ever apply body exposure?" "Yes"/ "No"). Moreover, a few open answer questions (e.g., "Do you have any further comments on the survey") were included; see supplementary materials and Open Science Framework Project for more details. Additionally, we also included several questions about the specific implementation of BE (e.g., duration of BE sessions, use of homework), which are part of a different research question.

Procedure

The survey was online between November 2019 and August 2022. Upon accessing the survey, a brief definition of BE, study aims and eligibility criteria as well as a policy statement on confidentiality and data security were provided. Participants were informed that all responses were assessed, analysed, and stored anonymously and were then given the choice to provide informed consent online. Participants could access the survey only after they had provided their informed consent. As compensation for their time and effort, we offered an online BE workshop to the participants. Interested participants had then to indicate their email address in a separate survey accessible over a link in order to receive access to the workshop invitation. This separate link guaranteed the anonymous data collection of responses during the survey. The study protocol and survey questions were approved by the Ethical Committee of the Albert-Ludwigs University Freiburg, Germany (494/19).

Data analyses

Data were analysed with the statistical software program SPSS. If not indicated otherwise, mean values and standard deviations of Likert-scale items were calculated. Two questions about attitudes towards exposure in general and two questions about attitudes towards BE specifically were asked, using similar wording ("In general, I believe that (body) exposure is an important technique" and "I use (body) exposure regularly"). These questions were designed based on prior research by Becker et al. (2004), see supplementary materials for more details. A compound score of these four questions was computed comprising positive attitudes towards (body) exposure.

Negative beliefs towards body exposure were evaluated with 13 questions adapted from the German version of the therapist beliefs about exposure scale (TBES; Deacon et al. 2013; Schumacher et al. 2019) and a sum score was computed, see supplementary materials for more details. Missing values of these items were not imputed and cases with missing values were excluded from analyses on negative beliefs. Confidence of BE was measured based on the mean of two questions about knowledge and selfefficacy when using BE, see supplementary materials for more details.

When applicable (e.g., when comparing practitioners using BE with those who do not use BE) independent sample *t*-tests were used and degrees of freedom were adjusted if variances were unequal and are reported accordingly. When correlations of two categorical nominal variables were conducted (e.g., gender and use of body exposure), cross-tabulation correlations were conducted and X^2 (Chi-Square) statistics were reported, otherwise Pearson's correlation coefficient was reported.

Results

Due to the unknown number of people who received the invitation email, it was not possible to calculate a precise response rate. Overall, 1.132 people accessed the survey link and 668 people (59%) started the survey, but discontinued after the first page which stated the purpose of the survey. Overall, 283 people (25%) completed

Table 1 Characteristics of participating psychotherapists

Characteristics		Overall sample (N=230)	BE-therapists (n=134)	Non-BE therapists (n=96)	Statistics ^g
Age (M, SD)		45.31 (9.43) ^a	45.18 (9.13)	45.50 (9.88)	t(227)=0.25
Gender (f/m/non-binary)		190/35/4 ^a	111/19/3 ^b	79/16/1	$X^2(df=3, N=230)=1.41$
Psychotherapy Training					$X^2(df = 1, N = 230) = 27.34^{**}$
	CBT	180	121	59	
	PD	50	13	37	
Specialisation					$X^2(df = 2, N = 230) = 4.20$
	Psychological Psychotherapist for Chil- dren & Adolescents	56	32	24	
	Psychological Psychotherapist for Adults	155	95	32	
	Medical doctor with psychotherapy training	19	7	12	
Years of Experience (M, SD)		11.77 (7.70)	12.51 (7.32)	10.74 (8.13)	t(228) = 1.72
% of ED patients (M, SD)		12.88 (12.23)	14.26 (13.13)	10.95 (10.62)	t (224.6) = 2.11*
Negative beliefs ^c (M, SD)		25.18 (5.26)	23.96 (4.57)	26.95 (5.70)	$t(220)^{d} = 4.32^{**}$
Positive attitudes ^e (<i>M</i> , <i>SD</i>)		12.21 (2.59)	13.10 (2.32)	10.97 (2.43)	$t(228) = 6.74^{**}$
Confidence ^f (<i>M</i> , <i>SD</i>)		2.05 (0.80)	2.44 (0.72)	1.51 (0.55)	$t(227.1) = 11.08^{**}$

Note. M=Mean, SD=Standard deviation, f=female, m=male, BE therapists=therapists who indicated to have applied BE; non-BE therapists=therapists who reported to have not applied BE.^a based on n=229, because one answer was missing.^b based on n=133, because one answer was missing.^c based on a sum score comprising 13 questions of the BE-adapted German Version of the Therapist Beliefs Scale (Deacon et al. 2013; German Version see Schumacher et al. 2019) range: 13–52, higher scores = more negative beliefs, ^d based on n = 222, because eight participants had missing values.^e based on the sum score comprising four statements on the use of BE (two questions) and exposure in general (two questions), range 4–16, higher scores = more positive attitudes. ^f based on the average score of two questions on knowledge and security when using BE (range 1–4, higher scores = more confidence).^g Statistics refer to a comparison between BE and non-BE therapists, the statistical test applied is indicated in the respective cell.^{*} Significant at p <.05, ^{**} significant at p <.001

the survey. Of the 283 participants who completed the survey, 21 were not licensed psychotherapists and 26 indicated they were medical doctors without formal psychotherapy training, therefore these participants (n = 47)were excluded from further analyses. Of the remaining participants (n = 236), five therapists indicated that they were trained in other therapy approaches than cognitive behavioural therapy and psychodynamic therapies. As these numbers were too small for meaningful separate analyses (e.g., concerning the association with usage of BE), we chose to exclude those five therapists from further analyses. One participant who indicated to use BE was excluded, as their description of the applied BE procedure was not in concordance with usual procedures. Overall, this resulted in a final sample of 230 licensed psychotherapists for the analyses.

A total of 155 of the eligible 230 participants were psychotherapists for adults (67.4%), 56 worked as child - and adolescent psychotherapists (24.4%) (10 of those worked as child and adolescents as well as adult psychotherapists) and 19 worked as medical doctor with psychotherapy training (8.3%). The majority of participating psychotherapists were cognitive behavioural therapists (78.3%) and 21.7% were psychodynamic psychotherapists. Most participating psychotherapists were women (82.6%; 0.4% specified as non-binary) and mean age was 45.3 years (SD = 9.4, Range = 24-72), see also Table 1 for sample characteristics.

On average, participating clinicians reported a working experience of 11.8 years (SD = 7.7, Range = 0-45) with EDs and that patients with EDs made up about 12.9% (SD = 12.2, Range = 0 - 80%) of their total patient load. Most clinicians indicated that they had seen about one to five patients of each ED diagnosis category: Anorexia Nervosa (AN; 48.7% of therapists), Bulimia Nervosa (BN; 45.7% of therapists), Binge Eating Disorder (BED; 38.7% of therapists) and Other Specified Feeding and Eating Disorders (OSFED; 40.9% of therapists). A minority of therapists indicated that they treated more than 20 patients with AN (11.7%), BN (11.7%), BED (10.0%) and OSFED (13.0%). On average, more than half (53.7%) of participating therapists indicated that they treated their ED patients in an outpatient (vs. inpatient) and individual (vs. group format) (65.5%) client setting.

Dissemination of body exposure in clinical practice

The majority of participants indicated that they have used BE in clinical practice (58.3%). Therapists who used BE delivered this technique, on average, to 12.4 patients (SD = 17.0, *Range* = 0–100) and treated 37.3% of their total ED patient load with BE.

The majority of therapists who used BE delivered BE specifically to patients with EDs (93.3%), particularly to patients with AN (75.4% indicated use of BE for AN) and BN (56.7% indicated use of BE for BN). BE was less often applied in the treatment of OSFED (40.3% indicated use

of BE for OSFED) and treatment of BED (20.9% therapists indicated use of BE for BED).

Additionally, 35.1% of therapists using BE indicated that they delivered BE to patients with Body Dysmorphic Disorder (BDD) and 21.6% used BE for patients with other psychological or medical problems (e.g., people with lower self-esteem, people with obesity, people with posttraumatic anxiety disorder and body dysphoria in transgender individuals).

The majority of therapists using BE (87.3%) indicated that they have also used other techniques to treat body image disturbances, such as cognitive interventions (81.3%), drawing of the body silhouette (59.0%) and estimating body size with a rope/tape measure (32.8%). Additionally, 15.7% reported to apply additional techniques such as mindfulness, body-related imagery exercises and body-related chair dialogues. Therapists who applied BE were also asked to indicate the percentage of using BE in proportion to all applied techniques to treat body image disturbances. Results yielded that BE made up, on average, 24.5% of all techniques used to treat body image disturbances (SD = 21.0; Range = 0–100). However, of note, 38.5% of therapists who did not use BE indicated that they do not apply any other techniques to treat body image disturbances. Thus, the answering pattern of this group of participants suggests that no specific treatment for body image disturbances is implemented during treatment.

Associations of psychotherapists' characteristics, attitudes and beliefs with the dissemination of body exposure

To evaluate if and how specific characteristics of clinicians relate to the dissemination of BE, we conducted correlational analyses of clinicians' demographic variables, experience-related variables (such as proportion of ED patients and type of psychotherapy training) and attitudes with the delivery of BE.

Age (r(229) = -0.02, p = .80) and gender $(X^2(df = 3, p = .80))$ N=230 = 1.41, p=.70) were not significantly associated with the use of BE (i.e., if BE was used: yes/no). Duration of working experience (in years) with EDs (r(230) = -0.11, p=.09) and professional specialisation (i.e., adult, child - and adolescents, or medical doctors with psychotherapy training $(X^2(df=2, N=230)=4.20, p=.12)$ yielded non-significant associations with the application of BE. Instead, a post-hoc correlation analysis yielded that being trained in BE was significantly associated with implementation of BE, $X^2(df=1, N=230) = 25.03$, p < .001: the majority of therapists who indicated that they were trained in BE also indicated that they implement BE in daily practice (82.5%). In contrast, half of therapists, who did not receive a training for BE (52.5%) indicated that they have never implemented BE in daily practice.

The use of BE also differed significantly depending on the type of psychotherapy training (i.e., cognitive-behavioural therapy, psychodynamic therapies): 67.2% of participating cognitive-behavioural therapists delivered BE, whereas 26% of psychodynamic therapists indicated to use this technique, $X^2(df=1, N=230)=27.34$, p<.001. Moreover, compared to therapists who never used BE, therapists using BE treated a significantly larger proportion of patients with EDs (14.3% vs. 10.9%) within their total patient load, see Table 1. Similarly, a significant correlation (r(230) = 0.13, p=.04) demonstrated the positive association of BE application and proportion of ED patients treated within total patient load.

Moreover, therapists who never used BE reported overall significantly more negative beliefs about (body) exposure and expressed less positive attitudes towards (body) exposure than therapists using BE. In addition, therapists who applied BE were significantly more confident in BE, see Table 1. Training in BE was also significantly associated with positive attitudes towards (body) exposure and confidence in BE, (r(230) = 0.25, p < .001 and r(230) = 0.40, p < .001, respectively). Training in BE was, however, not associated with negative beliefs about (body) exposure (r(230) = -0.05, p = .46).

Within the group of therapists who applied BE, confidence in BE correlated significantly positively with the number of patients treated with BE, r(134) = 0.51, p <.001 and the percentage of using BE in proportion to other techniques used to treat body image disturbances, r(134) = 0.46, p < .001. Results yielded that positive attitudes towards (body) exposure (based on the compound score) correlated positively with the proportion of patients treated with BE, r(134) = 0.27, p = .002. Negative beliefs about BE correlated negatively with the proportion of using BE as a technique to treat body image disturbances, r(131) = -0.28, p = .001. Within this group of therapists, negative beliefs about BE did not correlate significantly with age, years of working experience, or with the percentage of patients with eating disorder treated, r(131) = 0.12, p = .19, r(131) = -0.12, p = .18 and r(131) = -0.12, p = .18, respectively.

Evaluation of benefits and side-effects of body exposure in daily practice

Therapists using BE reported that, on average, the majority (62.1%) of their patients profit from BE. A range of several potential benefits and side-effects of BE were then rated on 4-point Likert scales (answering options were: "not applicable at all" (1), "not applicable" (2), "applicable" (3), "totally applicable" (4), see Open Science Framework (https://osf.io/fz9ys/) for the complete questionn aire and all answering options). Overall, therapists who applied BE rated an improved regulation of negative body-related emotions (M = 3.43, SD = 0.6), a reduction of biased body-related perception (M = 3.40, SD = 0.6) as well as a reduction in negative body-related cognitions (M = 3.40, SD = 0.6) as most important indicators of BE benefits. Moreover, within the group of therapists who implemented BE, confidence correlated significantly with the rate of patients who they reported improved, r(131) = -0.12, p = .18. In other words, based on these therapists' self-reports, BE was more effective when the therapist had a high confidence in BE r(134) = 0.37, p < .001.

As negative side effects, therapists who applied BE observed most often self-depreciation (M = 2.8, SD = 0.8) and an increased focus on negative cognitions and emotions (M = 2.4, SD = 0.8). Drop-out from therapy was rated as least often experienced side effect with 98.5% of therapists who applied BE indicated that they have never or rarely experienced drop-out from therapy due to BE. Only two therapists (out of 134 therapists who applied BE, i.e., 1.4%) indicated that they have regularly (n = 1) or often (n = 1) experienced drop-out due to BE. No other serious side effects were reported in a voluntary open answer format by therapists.

Discussion

The primary aim of this study was to quantify the dissemination of BE as an evidence-based technique for body image disturbances in patients with EDs in daily practice in Germany. In addition, we wanted to test whether therapist characteristics such as demographic background, clinical specialisation and attitudes towards exposure were associated with the uptake of BE. Moreover, we aimed to assess therapists' perception of the benefits and side-effects of BE in clinical practice.

Most importantly, the results of our online survey suggest that BE is relatively well disseminated in clinical practice in Germany: More than half of all participating therapists (58%) indicated that they have used this technique. The percentage of therapists offering BE is relatively high, compared to the prevalence of other (disorder-specific) exposure techniques offered in clinical practice. For example, studies on the dissemination of exposure in treatment of anxiety disorders often reported the use of (specific) exposure techniques in less than half of their samples (e.g., Becker-Haimes et al. 2017; Freiheit et al. 2004; Moses et al. 2021; Pittig and Hoyer 2017). Perhaps therapists treating EDs tend to specialise in EDspecific treatment techniques such as BE because EDs are challenging to treat (Schmidt et al. 2016). This is in line with our data showing that about 43% of therapists who used BE received a specific BE training (which is also a fairly high percentage compared to similar research, see e.g., Becker et al. 2004). Likewise, about 71% of participating therapists (203 out of 283) responded to our offer of a free online BE-training as compensation for their effort to complete the survey.

Although the majority of therapists reported using BE, it is important to note that BE was only delivered to about a third of all patients (i.e., 37%). This finding suggests that there is room for improvement in the dissemination of BE, especially considering that BE has clearly acquired the strongest evidence base as a treatment technique for body image disturbances at this point (Alleva et al. 2015; Hartmann et al. 2021; Naumann et al. 2022). Although 87.3% of therapists using BE reported also using additional techniques (mainly cognitive interventions) to treat body image disturbances, 38.5% of therapists who did not use BE reported also not using any additional technique targeted at body image disturbances. This finding is concerning given that body image disturbances are a core symptom of EDs and pose a risk of relapse if left untreated (Glashouwer et al. 2019; Jacobi et al. 2004).

Although the survey-completion rate of 25% was similar to previous research on exposure provision in clinical practice in Germany (e.g., 28.6% in Pittig and Hoyer 2017; 25.5% in Schumacher et al. 2019), one explanation for the comparably high percentage of BE dissemination may be that a selection bias occurred in the current study. During recruitment, we took active steps to avoid selection bias (e.g., by emailing all registered therapists and clinics, regardless of specialisation and professional background, and by emphasizing that therapists were invited to participate regardless of their experience with BE). However, it is possible that therapists who were familiar with the delivery of BE were more interested in supporting this research and therefore completed the survey, leading to an overestimation of the prevalence of usage of BE in clinical practice. The results should therefore be considered with this potential limitation in mind. Furthermore, the offer of a free online BE workshop as incentive for participation may also have contributed to a selection bias of therapists who were interested in BE. Future studies may consider other incentive methods (e.g., monetary compensation) to reduce this potential bias.

With regard to our secondary aim (i.e., the association between therapist characteristics and the use of BE), we found that the use of BE was more common among cognitive behavioural therapists than psychodynamic therapists— a finding that may in itself be explained by the different underlying theoretical conceptions of the therapeutic process. This is consistent with previous research showing that clinical orientation determines fundamental differences in ideas about treatment approaches (i.e., symptom-orientated cognitive behavioural approach vs. psychodynamic approaches see von Ranson and Robinson, 2006). Several previous (but not all) studies indicated that exposure is more often used by cognitive behavioural therapists than by therapists of other clinical orientations (e.g., de Jong et al. 2020; Langthorne et al. 2023).

Notably, our results indicated that therapists who never applied BE endorsed more negative beliefs about BE and held less positive attitudes towards exposure in general and BE in particular. This is consistent with numerous studies showing that negative beliefs about exposure hinder the implementation of exposure in clinical practice, despite overwhelming empirical evidence to the contrary (de Jong et al. 2020; Deacon et al. 2013; Langthorne et al. 2023). In contrast, positive attitudes towards BE and selfreported confidence in delivering BE were significantly associated with higher use of this therapeutic technique. These findings highlight the potential link between positive attitudes and the use of BE. These results also suggest that increasing confidence in BE delivery may be associated with greater use which was also found in a study on exposure training for therapists (Harned et al. 2013). Note, however, that additional interventions may be needed to increase the implementation of exposure techniques in daily practice, for example interventions using implementation intentions or interventions reducing concerns and anxiety about exposure use (Farrell et al. 2016; Schumacher et al. 2015; Trivasse et al. 2020).

In the current sample, therapists' age and gender did not correlate with the use of BE. This is in contrast to the results of previous research which found that female clinicians and older clinicians were less likely to use exposure in their daily practice (de Jong et al. 2020; Deacon et al. 2013). A possible explanation for this divergent finding may be that all participating therapists were licensed psychotherapists, as per the inclusion criterion, who may hold less negative beliefs about exposure in general and may therefore may be more likely to use exposure (Becker-Haimes et al. 2017; Deacon et al. 2013; Kline et al. 2021). This is in line with reports of another study among German therapists, which observed fewer reservations about exposure than generally found in comparable studies in the US (Schumacher et al. 2019). Notably, cognitive behavioural psychotherapy training in Germany often includes many opportunities to learn and to train exposure, which may therefore lead to fewer reservations about exposure and therefore to a higher overall delivery rate. Another explanation could be that BE is more commonly delivered by female therapists, as the majority of patients with EDs are female and prefer female therapists (Vocks et al. 2007). This may have affected our data, as the majority of participating therapists were female.

In terms of therapists' perceptions of the benefits and side-effects of BE, our results showed that BE was overall perceived to be effective: therapists using BE reported positive effects of BE for >60% of their patients. Moreover, confidence in BE correlated significantly positively with the percentage of patients who improved through BE. Thus, therapists who were more confident in BE, indicated that a higher number of their patients profited from BE. This result may indicate that boosting confidence of therapists in BE delivery, for example through training, may amplify treatment effects. This interpretation is in line with previous research showing a positive effect of therapists' confidence and competence on treatment outcome and treatment motivation (Bartle-Haring et al. 2022; Seewald and Rief 2024).

Therapists who applied BE described the benefits of BE as the regulation of body-related negative emotions and cognitions as well as the reduction of distorted body image as equally important determinants of treatment success. This observation by therapists converges with research showing significant improvements in negative body-related emotions and biased attentional processing of body parts in response to BE (Griffen et al. 2018; Hartmann et al. 2021; Krohmer et al. 2022; Naumann et al. 2022). Thus, the benefits of BE in controlled settings appear to be consistent with what is observed in clinical practice. Note, however, that this data is based on therapists' evaluation of benefits and not on clients' self-report in the current study. Moreover, we did not assess how therapists came to this evaluation (e.g., based on clinical impression, patient self-report or questionnaires).

Furthermore, in the current study, therapists who applied BE, indicated that negative side-effects were rarely experienced in clinical practice. This is particularly relevant as a previous review of the adverse effects of BE highlighted the inconclusive evidence on the adverse effects of BE (Griffen et al. 2018). Moreover, this finding may also indicate that clinicians' fears of imposing a heavy emotional distress on patients when using exposure-based techniques, such as BE, are not consistent with clinical experience in daily practice as reported by participants in this study. Instead, this finding should encourage clinicians to use BE despite their potential inhibiting fear of not being sure about the outcome of a BE session (see for a similar argumentation Mulkens et al. 2018). Using BE as often as possible will increase confidence in this technique, which will in turn increase the use of BE, as our results suggest. Training in BE and supervision may further facilitate this effect (Michael et al. 2021; Moses et al. 2023).

Overall, we conclude that even though the majority of therapists in our sample were familiar with BE, BE is applied in less than half of all treated patients. This is particularly important against the background that BE may be currently the best technique available to treat body dissatisfaction (Alleva et al. 2015), at least within a cognitive behavioural therapy-treatment context, and a substantial number of patients may not receive any interventions for body image disturbances. Thus, these results call for continued development to foster optimal treatment of EDs in clinical practice.

One way to increase the uptake of BE within cognitive behavioural therapy treatment is through training and supervision opportunities. Our findings highlight the importance of training in BE with the uptake of BE, attitudes towards BE and confidence in BE. This is in line with previous research that emphasized the potential of addressing positive attitudes towards exposure directly during therapy training in order to increase the dissemination of exposure in clinical practice (Deacon et al. 2013; Farrell et al. 2016; Simmons et al. 2008; Trivasse et al. 2020). However, note, that in a survey among German therapists, more training and/or supervision opportunities were named as least important factors influencing the decision to apply exposure for patients with anxiety disorders (Pittig and Hoyer 2017). In that specific study, environment-related factors (such as uncertainty about remuneration and insurance) were named as most important for the (non)-utilization of exposure. Moreover, Schumacher and her colleagues (Schumacher et al. 2014, 2015) found that exposure was linked to high stress responses in clinicians during delivery, which may impact the likelihood and caution of exposure delivery (Deacon et al. 2013; Langthorne et al. 2023). Thus, it is important for future research to determine which specific factors can encourage and can motivate psychotherapists to implement BE to the majority of their patients with body image disturbances. Similarly, it is important to gain a better understanding of barriers that impede therapists from using BE.

In conclusion, our survey revealed that therapists working with ED perceive BE as a beneficial technique for addressing body image disturbances in clinical practice. However, the finding that BE is only applied in about a quarter of all ED cases, and that confidence and positive attitudes are significantly associated with the use of BE highlights the need to lower therapists' individual barriers to use this disorder-specific exposure technique.

Abbreviations

AN	Anorexia Nervosa	
BE	Body Exposure	
BED	Binge Eating Disorder	
BDD	Body Dysmorphic Disorder	
BN	Bulimia Nervosa	
ED	Eating Disorder	

OSFED Other specific feeding and eating disorders

Supplementary Information

The online version contains supplementary material available at https://doi.or g/10.1186/s40337-025-01262-6.

Supplementary Material 1

Page 8 of 10

Acknowledgements

We thank all participating psychotherapists for their time and effort in completing our survey.

Author contributions

All authors conceptualized the study. JW and ASH were involved in recruitment and testing. JW analysed the data. JW and ASH wrote the first draft of the manuscript. All authors contributed to the rewriting of the manuscript and agreed to the final version.

Funding

Open Access funding enabled and organized by Projekt DEAL. This study was funded by a DFG Network Grant (HA 8589/2 – 1) awarded to ASH.

Data availability

The online survey (in German language) and the dataset with relevant variables analysed during the current study are available in the Open Science Framework repository, osf.io/fz9ys/.

Declarations

Ethics approval and consent to participate

This study was approved by the Ethical Committee of the Albert-Ludwigs University Freiburg, Germany (494/19).

Consent for publication

Not applicable.

Competing interests

The authors declare no competing interests.

Author details

¹Institute of Psychology, Department of Clinical Psychology and Psychotherapy, Albert-Ludwigs-University Freiburg, Freiburg, Germany ²Institute of Psychology, Department of Clinical Psychology and Psychotherapy, Eberhard Karls University Tübingen, Tübingen, Germany ³Institute of Psychology, Department of Clinical Psychology and Psychotherapy, University of Osnabrück, Osnabrück, Germany ⁴Department of Psychology, Unit of Clinical Psychology and Psychotherapy of Childhood and Adolescence, University of Konstanz, Konstanz, Germany

Received: 8 September 2024 / Accepted: 8 April 2025 Published online: 18 April 2025

References

- Alleva JM, Sheeran P, Webb TL, Martijn C, Miles E. A meta-analytic review of standalone interventions to improve body image. PLoS ONE. 2015;10(9):1–32. http s://doi.org/10.1371/journal.pone.0139177.
- American Psychiatric Association. (2013). Diagnostic and statistical manual of mental disorders: DSM-5. In DSM-5.
- Arbeitsgemeinschaft der Wissenschaftlichen Medizinischen Fachgesellschaften. (2019). 53-Leitlinien: Diagnostik und Therapie der Essstörungen. https://www.aw mf.org/leitlinien/detail/II/051-026.html
- Bartle-Haring S, Bryant A, Whiting R. Therapists' confidence in their theory of change and outcomes. J Marital Fam Ther. 2022;48(4):1190–205. https://doi.o rg/10.1111/jmft.12593.
- Baur J, Krohmer K, Naumann E, Svaldi J. Efficacy and mechanisms of change in exposure-based and cognitive stand-alone body image interventions in women with overweight and obesity. Behav Res Ther. 2022;159(July):104210. https://doi.org/10.1016/j.brat.2022.104210.
- Becker CB, Zayfert C, Anderson E. A survey of psychologists' attitudes towards and utilization of exposure therapy for PTSD. Behav Res Ther. 2004. https://doi.org /10.1016/S0005-7967(03)00138-4.
- Becker-Haimes EM, Okamura KH, Wolk CB, Rubin R, Evans AC, Beidas RS. Predictors of clinician use of exposure therapy in community mental health settings. J Anxiety Disord. 2017;49(April):88–94. https://doi.org/10.1016/j.janxdis.2017. 04.002.

- Biney H, Astbury S, Haines A, Grant J, Malone N, Hutt M, Matthews R, Morgan JF, White S, Lacey JH. A novel 'practical body image' therapy for adolescent inpatients with anorexia nervosa: a randomised controlled trial. Eat Weight Disorders. 2021;26(6):1825–34. https://doi.org/10.1007/s40519-020-00997-2.
- Cook JM, Schnurr PP, Foa EB. Bridging the gap between posttraumatic stress disorder research and clinical practice: the example of exposure therapy. Psy-chotherapy. 2004;41(4):374–87. https://doi.org/10.1037/0033-3204.41.4.374.
- Cook JM, Biyanova T, Elhai J, Schnurr PP, Coyne JC. What do psychotherapists really do in practice? An internet study of over 2,000 practitioners. Psychotherapy. 2010;47(2):260–7. https://doi.org/10.1037/a0019788.
- de Jong R, Lommen MJJ, van Hout WJPJ, de Jong PJ, Nauta MH. Therapists' characteristics associated with the (non-)use of exposure in the treatment of anxiety disorders in youth: A survey among Dutch-speaking mental health practitioners. J Anxiety Disord. 2020;73(May):102230. https://doi.org/10.1016 /j.janxdis.2020.102230.
- Deacon BJ, Farrell NR, Kemp JJ, Dixon LJ, Sy JT, Zhang AR, McGrath PB. Assessing therapist reservations about exposure therapy for anxiety disorders: the therapist beliefs about exposure scale. J Anxiety Disord. 2013;27(8):772–80. ht tps://doi.org/10.1016/j.janxdis.2013.04.006.
- Díaz-Ferrer S, Rodríguez-Ruiz S, Ortega-Roldán B, Moreno-Domínguez S, Fernández-Santaella MC. Testing the efficacy of pure versus guided mirror exposure in women with bulimia nervosa: A combination of neuroendocrine and psychological indices. J Behav Ther Exp Psychiatry. 2015;48:1–8. https://d oi.org/10.1016/j.jbtep.2015.01.003.
- Fairburn CG, Cooper Z, Shafran R. (2003). Cognitive behaviour therapy for eating disorders: A transdiagnostic theory and treatment. In *Behaviour Research and Therapy* (Vol. 41, Issue 5, pp. 509–528). https://doi.org/10.1016/S0005-7967(02)00088-8
- Farrell NR, Kemp JJ, Blakey SM, Meyer JM, Deacon BJ. Targeting clinician concerns about exposure therapy: A pilot study comparing standard vs. enhanced training. Behav Res Ther. 2016;85:53–9. https://doi.org/10.1016/j.brat.2016.0 8.011.
- Finch J, Ford C, Grainger L, Meiser-Stedman R. A systematic review of the clinician related barriers and facilitators to the use of evidence-informed interventions for post traumatic stress. J Affect Disord. 2020;263(November 2019):175–86. h ttps://doi.org/10.1016/j.jad.2019.11.143.
- Freiheit S, Vye C, Swan R, Cady M. Cognitive-Behavioral therapy for anxiety: is dissemination working?? Behav Therapist. 2004;27(2):25–32.
- Glashouwer KA, Jonker NC, Thomassen K, de Jong PJ. Take a look at the bright side: effects of positive body exposure on selective visual attention in women with high body dissatisfaction. Behav Res Ther. 2016;83:19–25. https://doi.org/10.1 016/j.brat.2016.05.006.
- Glashouwer KA, van der Veer RML, Adipatria F, de Jong PJ, Vocks S. The role of body image disturbance in the onset, maintenance, and relapse of anorexia nervosa: A systematic review. Clin Psychol Rev. 2019;74(October 2018):101771. ht tps://doi.org/10.1016/j.cpr.2019.101771.
- Griffen TC, Naumann E, Hildebrandt T. Mirror exposure therapy for body image disturbances and eating disorders: A review. Clin Psychol Rev. 2018;65:163–74. h ttps://doi.org/10.1016/j.cpr.2018.08.006.
- Harned MS, Dimeff LA, Woodcock EA, Contreras I. Predicting adoption of exposure therapy in a randomized controlled dissemination trial. J Anxiety Disord. 2013;27(8):754–62. https://doi.org/10.1016/j.janxdis.2013.02.006.
- Hartmann AS, Naumann E, Vocks S, Svaldi J, Werthmann J. (2021). Body Exposure, its Forms of Delivery and Potentially Associated Working Mechanisms: How to Move the Field Forward. In *Clinical Psychology in Europe* (Vol. 3, Issue 3). htt ps://doi.org/10.32872/cpe.3813
- Hilbert A, Tuschen-Caffier B. Body image interventions in cognitive-behavioural therapy of binge-eating disorder: A component analysis. Behav Res Ther. 2004;42(11):1325–39. https://doi.org/10.1016/j.brat.2003.09.001.
- Jacobi C, Hayward C, de Zwaan M, Kraemer HC, Agras WS. Coming to terms with risk factors for eating disorders: application of risk terminology and suggestions for a general taxonomy. Psychol Bull. 2004;130(1):19–65. https://doi.org/ 10.1037/0033-2909.130.1.19.
- Kline AC, Klein AB, Bowling AR, Feeny NC. Exposure therapy beliefs and utilization for treatment of PTSD: A survey of licensed mental health providers. Behav Ther. 2021;52(4):1019–30. https://doi.org/10.1016/j.beth.2021.01.002.
- Krohmer K, Naumann E, Tuschen-Caffier B, Svaldi J. Mirror exposure in binge-eating disorder: changes in eating pathology and attentional biases. J Consult Clin Psychol. 2022;90(8):613–25. https://doi.org/10.1037/ccp0000751.
- Langthorne D, Beard J, Waller G. Therapist factors associated with intent to use exposure therapy: a systematic review and meta-analysis. Cogn Behav Ther. 2023;52(4):347–79. https://doi.org/10.1080/16506073.2023.2191824.

- Legenbauer T, Vocks S. (2014). Manual der kognitiven Verhaltenstherapie bei Anorexie und Bulimie (2. Auflage). Springer-Verlag. https://doi.org/10.1007/978-3-64 2-20385-5
- Lewer M, Kosfelder J, Michalak J, Schroeder D, Nasrawi N, Vocks S. Effects of a cognitive-behavioral exposure-based body image therapy for overweight females with binge eating disorder: a pilot study. J Eat Disorders. 2017;5:43. ht tps://doi.org/10.1186/s40337-017-0174-y.
- Michael ST, Blakey SM, Sherrill AM, Aosved AC. Exposure therapy training and supervision: Research-informed strategies for addressing barriers to adoption and dissemination. Train Educ Prof Psychol. 2021;15(2):97–105. https://doi.org /10.1037/tep0000304.
- Milgram L, Freeman JB, Benito KG, Elwy AR, Frank HE. Clinician-Reported determinants of Evidence-Based practice use in private practice mental health. J Contemp Psychother. 2022;52:337–46. https://doi.org/10.1007/s10879-022-0 9551-w.
- Moreno-Domínguez S, Rodríguez-Ruiz S, Fernández-Santaella MC, Jansen A, Tuschen-Caffier B. Pure versus guided mirror exposure to reduce body dissatisfaction: A preliminary study with university women. Body Image. 2012;9(2):285–8. https://doi.org/10.1016/j.bodyim.2011.12.001.
- Moses K, Gonsalvez C, Meade T. Utilisation and predictors of use of exposure therapy in the treatment of anxiety, OCD and PTSD in an Australian sample: a preliminary investigation. BMC Psychol. 2021;9(1):1–11. https://doi.org/10.118 6/s40359-021-00613-7.
- Moses K, Gonsalvez CJ, Meade T. Barriers to the use of exposure therapy by psychologists treating anxiety, obsessive-compuslive disorder, and posttraumatic stress disorder in an Australian sample. J Clin Psychol. 2023;79(4):1156– 65. https://doi.org/10.1002/jclp.23470.
- Mulkens S, de Vos C, de Graaff A, Waller G. To deliver or not to deliver cognitive behavioral therapy for eating disorders: replication and extension of our Understanding of why therapists fail to do what they should do. Behav Res Ther. 2018;106(2018):57–63. https://doi.org/10.1016/j.brat.2018.05.004.
- National Institute for Health and Care Excellence (NICE). Eating disorders: recognition and treatment. National Institute for Health and Care Excellence (NICE); 2017.
- Naumann E, Werthmann J, Vocks S, Svaldi J, Hartmann AS. Die spiegelkonfrontationstherapie Zur behandlung von Körperbildstörungen Bei Essstörungen– Evidenz, Wirkmechanismen und vorgehensweise. Psychologische Rundschau. 2022;73(4):243–59. https://doi.org/10.1026/0033-3042/a000558.
- Pittig A, Hoyer J. Exposition Aus sicht niedergelassener verhaltenstherapeutinnen und verhaltenstherapeuten. Z Für Klinische Psychologie Und Psychother. 2017;46(4):223–35. https://doi.org/10.1026/1616-3443/a000441.
- Schmidt U, Adan R, Böhm I, Campbell IC, Dingemans A, Ehrlich S, Elzakkers I, Favaro A, Giel K, Harrison A, Himmerich H, Hoek HW, Herpertz-Dahlmann B, Kas MJ, Seitz J, Smeets P, Sternheim L, Tenconi E, van Elburg A, Zipfel S. Eating disorders: the big issue. Lancet Psychiatry. 2016;3(4):313–5.
- Schumacher S, Gaudlitz K, Plag J, Miller R, Kirschbaum C, Fehm L, Fydrich T, Ströhle A. Who is stressed? A pilot study of salivary cortisol and alpha-amylase concentrations in agoraphobic patients and their novice therapists undergoing in vivo exposure. Psychoneuroendocrinology. 2014;49(1):280–9. https://doi.or g/10.1016/j.psyneuen.2014.07.016.
- Schumacher S, Miller R, Fehm L, Kirschbaum C, Fydrich T, Ströhle A. Therapists' and patients' stress responses during graduated versus flooding in vivo exposure in the treatment of specific phobia: A preliminary observational study. Psychiatry Res. 2015;230(2):668–75. https://doi.org/10.1016/j.psychres.2015. 10.020.
- Schumacher S, Schopka NM, Heinrich M, Knaevelsrud C. The German adaptation of the therapist beliefs about exposure scale: A validation study among licensed cognitive behavioural therapists in Germany. Behav Cogn Psychother. 2019;47(2):164–80. https://doi.org/10.1017/S1352465818000371.
- Seewald A, Rief W. Therapist's warmth and competence increased positive outcome expectations and alliance in an analogue experiment. Psychother Res. 2024;34(5):663–78. https://doi.org/10.1080/10503307.2023.2241630.
- Simmons AM, Milnes SM, Anderson DA. Factors influencing the utilization of empirically supported treatments for eating disorders. Eat Disord. 2008;16(4):342–54. https://doi.org/10.1080/10640260802116017.
- Stice E, Shaw HE. Role of body dissatisfaction in the onset and maintenance of eating pathology: A synthesis of research findings. J Psychosom Res. 2002;53(5):985–93. https://doi.org/10.1016/S0022-3999(02)00488-9.
- Svaldi J, Tuschen-Caffier B. Bulimia nervosa (Reihe: For). Hogrefe Verlag GmbH & Co KG; 2018.
- Tivian XIGH. (2021). UniPark. www.unipark.com.

- Trentowska M, Bender C, Tuschen-Caffier B. Mirror exposure in women with bulimic symptoms: how do thoughts and emotions change in body image treatment? TT - Spiegelexposition Bei Frauen Mit Bulimischen symptomen: Wie verändern Sich gedanken und gefühle Bei einer körperbildbehandlung?? Behav Res Ther. 2013;51(1):1–6. https://doi.org/10.1016/j.brat.2012.03.012.
- Trentowska M, Svaldi J, Blechert J, Tuschen-Caffier B. Does habituation really happen? Investigation of psycho-biological responses to body exposure in bulimia nervosa. Behav Res Ther. 2017;90:111–22. https://doi.org/10.1016/j.b rat.2016.12.006.
- Trivasse H, Webb TL, Waller G. A meta-analysis of the effects of training clinicians in exposure therapy on knowledge, attitudes, intentions, and behavior. Clin Psychol Rev. 2020;80:101887. https://doi.org/10.1016/j.cpr.2020.101887.
- Tuschen-Caffier B, Florin I. Teufelskreis bulimie. Ein manual Zur Psychologischen therapie. Hogrefe Verlag GmbH & Co KG; 2012.
- Vocks S, Legenbauer T, Peters I. Does shape matter? Preferences for a female therapist's figure among patients with eating disorders. Psychother Res. 2007a;17(4):416–22. https://doi.org/10.1080/10503300600740703.
- Vocks S, Legenbauer T, Wächter A, Wucherer M, Kosfelder J. What happens in the course of body exposure? Emotional, cognitive, and physiological

reactions to mirror confrontation in eating disorders. J Psychosom Res. 2007b;62(2):231–9. https://doi.org/10.1016/j.jpsychores.2006.08.007.

- Vocks S, Wächter A, Wucherer M, Kosfelder J. Look at yourself: can body image therapy affect the cognitive and emotional response to seeing oneself in the mirror in eating disorders? Eur Eat Disorders Rev. 2008;16(2):147–54. https://d oi.org/10.1002/erv.825.
- Vocks S, Bauer A, Legenbauer T. (2018). Körperbildtherapie bei Anorexia und Bulimia nervosa Ein kognitiv-verhaltenstherapeutisches Behandlungsprogramm (3. Auflage). Hogrefe Verlag GmbH & Co KG.
- Wright C, Waller G. The impact of teaching clinicians about implementing exposure therapy with patients with eating disorders: A nonrandomized controlled study. Int J Eat Disord. 2020;53(1):107–12. https://doi.org/10.1002 /eat.23171.

Publisher's note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.